Author's response to reviews

Title: Identifying nurses’ rewards: a qualitative categorization study in Belgium

Authors:

Sara De Gieter (Sara.De.Gieter@vub.ac.be)
Rein De Cooman (Rein.De.Cooman@vub.ac.be)
Roland Pepermans (Roland.Pepermans@vub.ac.be)
Ralf Caers (Ralf.Caers@vub.ac.be)
Cindy Du Bois (Cindy.Du.Bois@vub.ac.be)
Marc Jegers (Marc.Jegers@vub.ac.be)

Version: 4 Date: 6 July 2006

Author's response to reviews:

We have corrected our manuscript (proofing stage) and made some small changes. Would it be possible to put Table 1 on one page (complete) instead of separating it? Thank you.

Kind regards,
Sara De Gieter